Author's response to reviews

Title: Sequelae following the use of the sural nerve for reconstructive surgery with nerve graft

Authors:

Alexander Hallgren (alexander.hallgren@gmail.com)
Anders Björkman (anders.bjorkman@med.lu.se)
Anette Chemnitz (anette.chemnitz@med.lu.se)
Lars B Dahlin (lars.dahlin@med.lu.se)

Version: 2 Date: 23 November 2012

Author's response to reviews:

Referring to E-mail contact with the editorial office for BMC (Miss Nikka Portodo). Miss Portodo recommended transfer of the manuscript from BMC Neuroscience to BMC Surgery.

Sincerely yours

Lars B. Dahlin